Previous Cesarean Section and the Risk of Preeclampsia: A Meta-analysis



Ensiyeh Jenabi¹ (), Salman Khazaei² (), Soodabeh Aghababaei^{3*} ()

- 1. Assistant Professor, Mother and Child Care Research Center, Hamadan University of Medical Sciences, Hamadan, Iran.
- 2. Assistant Professor, School of Health, Autism Spectrum Disorders Research Center, Hamadan University of Medical Sciences, Hamadan, Iran.
- 3. Associate Professor, School of Nursing and Midwifery, Mother and Child Care Research Center, Hamadan University of Medical Sciences, Hamadan, Iran.



Citation Jenabi E, Khazaei S, Aghababaei S. Previous Cesarean Section and the Risk of Preeclampsia: A Meta-analysis. J Holist Nurs Midwifery. 2023; 33(3):176-182. https://doi.org/10.32598/jhnm.33.3.2395

Running Title Previous Cesarean Section and Preeclampsia

doj https://doi.org/10.32598/jhnm.33.3.2395

\odot \odot

Article info: Received: 04/02/2022 Accepted: 29/01/2023 Available Online: 01/07/2023

ABSTRACT

Introduction: Preeclampsia is a common pregnancy complication with the multisystem variable disorder. Yet, the literature has not been systematically reviewed for the relationship between previous cesarean section and the risk of preeclampsia.

Objective: This study aimed to identify the relationship between previous cesarean delivery and the risk of preeclampsia.

Materials and Methods: This study was a systematic review and meta-analysis. PubMed, Scopus, ProQuest, and Web of Sciences were searched to identify eligible observational studies until May 25, 2019. The odds ratio (OR) and 95% confidence intervals (CI) were calculated as random effect estimates of association among studies. The quality of the included studies was examined based on the Newcastle-Ottawa scale.

Results: This study included 7 eligible articles (2 studies with a case-control design, 4 with a cohort design, and 1 with a cross-sectional design). The meta-analysis results showed an increased risk of preeclampsia in the women with previous cesarean section compared to women without cesarean section (OR=1.28, 95% Cl, 1.15%-1.41%, P=0.001), I²=37.2%. The quality of all studies except one study was high based on the Newcastle-Ottawa scale. The subgroup analysis was conducted based on the adjusted form of studies. The crude and adjusted studies were 1.29 (95% Cl, 0.13%-2.46%, P=0.2) and 1.29 (95% Cl, 1.22%-1.36%, P=0.001), respectively.

Keywords:

Cesarean section, Preeclampsia, Meta-analysis **Conclusion:** These findings showed that previous cesarean section is a risk factor for preeclampsia. Therefore, education programs and interventions should be considered to reduce elective cesarean section on maternal requests.

* Corresponding Author:

Soodabeh Aghababaei, Associate Professor.

Address: School of Nursing and Midwifery, Mother and Child Care Research Center, Hamadan University of Medical Sciences, Hamadan, Iran. Tel: +98 (81) 38380496

.....

E-mail: aghababaii@yahoo.com

Highlights

- Preeclampsia is a common complication of pregnancy with the multisystem variable disorder.
- Preeclampsia can cause maternal and fetal morbidity and death by affecting multiple organ systems.
- The results of studies have shown the relationship between previous cesarean delivery and the risk of preeclampsia.
- This meta-analysis determined the association between previous cesarean section and preeclampsia.

Plain Language Summary

Preeclampsia is a common pregnancy complication characterized by new-onset hypertension, edema, and proteinuria during pregnancy or immediately after delivery. Preeclampsia can cause maternal and fetal morbidity and death by affecting multiple organ systems. Cesarean-induced scar in the uterus can affect placenta implantation in a subsequent pregnancy. Abnormal uteroplacental blood flow and unusual trophoblastic invasion due to uterine changes in previous cesarean section may result in preeclampsia in subsequent pregnancies. This study aimed to identify the relationship between previous cesarean delivery and the risk of preeclampsia. We searched terms in PubMed, Scopus, ProQuest, Web of Sciences, and Google Scholar databases to identify eligible studies until May 25, 2019. This meta-analysis study showed an association between previous cesarean section and preeclampsia. These findings indicated that previous cesarean section is a risk factor for preeclampsia. We found a 28% increased risk for preeclampsia in women with previous cesarean section. This issue should be considered, especially for pregnant women who undergo cesarean delivery based on their request without any maternal or fetal indication.

Introduction

reeclampsia is a common pregnancy complication with the multisystem variable disorder, characterized by new-onset hypertension, edema, and proteinuria during pregnancy or immediately postpartum [1-4]. It affects about 2% to 8% of pregnancies, and in the second half of pregnancy, it is responsible for over 70000 maternal deaths and 500000 fetal deaths worldwide every year [4]. A blood pressure measurement greater than 140/90 mm Hg on two separate loads and urinary protein excretion exceeding 300 mg/d are clinical signs of clinical diagnosis [5]. The etiology of preeclampsia is complicated, and factors of the maternal and fetal and or paternal genetic determinants have the main role [6].

Preeclampsia can cause maternal and fetal morbidity and death by affecting multiple organ systems [7-9]. The risk of preeclampsia is increased in women with a previous history of preeclampsia, chronic and gestational diabetes, parity, multiple pregnancies, nulliparity, family history, body mass index before pregnancy, maternal age \geq 40 years, autoimmune disease, renal disease, chronic hypertension, and angiogenic factors [7, 10, 11]. Delivery can reduce most signs and symptoms of preeclampsia. In addition, preeclampsia can persist after delivery and develop during the postpartum period. In the peripartum period, preeclampsia is related to an increased risk of peripartum cardiomyopathy that can progress to chronic heart failure, cardiac transplantation, or death [12].

Evidence suggests that previous uterine scars can lead to placental-related complications such as abnormal placental abruption, preeclampsia, and preterm delivery [13-15]. It also appears that the initial pathology in preeclampsia is associated with maternal-fetal problems [16]. It is acknowledged that cesarean delivery is the most common surgery in obstetrics, and its rate continues to rise worldwide [17, 18]. Cesarean-induced scar in the uterus can affect placenta implantation in a subsequent pregnancy [19]. Abnormal uteroplacental blood flow and unusual trophoblastic invasion due to uterine changes in previous cesarean section may result in preeclampsia in subsequent pregnancies [20]. The results of studies have shown the relationship between previous cesarean delivery and the risk of preeclampsia [14, 20, 21].

Yet, the literature has not been systematically reviewed the relationship between previous cesarean section and the risk of preeclampsia. Therefore, the present metaanalysis study was designed to identify the relationship between previous cesarean delivery and the risk of preeclampsia.

Materials and Methods

This study was a systematic review and meta-analysis. We utilized a modified form of the preferred reporting items for systematic reviews and meta-analyses (PRIS-MA) checklist as a guide to enhance the quality reporting of the review. The literature search was conducted by the following keywords: (Previous cesarean section OR prior cesarean section OR previous cesarean OR prior cesarean) AND (preeclampsia OR preeclampsia OR pregnancy toxemia OR edema-proteinuria-hypertension gestosis) and based on MeSH database. The major international databases PubMed, Scopus, ProQuest, Web of Sciences, and Google Scholar search engine were searched to identify eligible studies from 1987 until May 24, 2022. We also searched gray literature, conferences, and references in the retrieved studies to find studies.

All studies that determined the association between previous cesarean section and the risk of preeclampsia were considered for inclusion in the present meta-analysis. The inclusion criteria were observational studies (case-control, cohort, and cross-sectional) that explained the association between previous cesarean section and the risk of preeclampsia. Review studies, letters to the Editor, and case reports were excluded from our research.

Two independent authors (EJ and SA) reviewed the retrieved studies, and the following information was extracted: (1) name of the first author, (2) publication year and location of study conduction, (3) total sample size, (4) age of maternal, (5) estimated odds ratio/risk ratio and (95% CIs), and (6) adjustment status (crude/ adjusted). Any disagreements were resolved by discussion among researchers. The subgroup analysis was conducted based on the adjusted form of studies.

To assess the quality of the included studies, the present meta-analysis conducted the Newcastle-Ottawa scale (NOS). It has 8 items and uses a star system to assess the quality of studies. According to this scale, studies range between 0 and 9 stars, and scores of the papers were divided into low quality (<7 points) and high quality (\geq 7 points) [22]. Two investigators performed the quality assessment independently. Any disagreement was resolved by discussion between the authors. We used Stata software, version 14 for the management and analysis of data. In this study, we evaluated the association between previous cesarean section and the risk of preeclampsia based on the odds ratio (OR) and its 95% confidence interval. A random-effects model was used to analyze of data. We assessed heterogeneity across studies by using the I² statistic. The values were 75%, 50%, and 25% based on cut-off points for high, moderate, and low heterogeneity. Publication bias was evaluated by Begg and Egger tests.

Results

We included 1384 studies via electronic databases based on search strategy and 241 studies via checking reference lists. We did not find new studies via searching conference databases. There were 523 duplicate studies. We excluded 1088 studies after reading the title and abstract, and 14 studies were considered for reading full papers. In the end, 7 studies were included in the present meta-analysis (Figure 1). These studies were four cohort studies [15, 20, 23, 24], two casecontrol [21, 25], and one cross-sectional [26] involving 423940 participants. In this systematic review, 4 studies out of 7 were adjusted based on confounding variables. Three studies were performed in Asia, 2 in America, and 2 in Africa (Table 1). The effect of previous cesarean section exposure on preeclampsia was determined by OR (Figure 2). The findings of studies showed an increased risk of preeclampsia in women with previous cesarean section compared to women without cesarean section: (OR=1.28, 95% CI, 1.15%-1.41%, P=0.001). The heterogeneity was low (I²=37.2%). The quality of studies was assessed, according to the NOS. The quality of all studies except one study was high.

The subgroup analysis was conducted based on the adjusted form of studies. ORs in crude and adjusted studies were reported at 1.29 (95% CI, 0.13%-2.46%, P=0.2) and 1.29 (95% CI, 1.22%-1.36%, P=0.001), respectively. A significant association was found in crude studies, and the heterogeneity among adjusted studies was low (I^2 =2.4%) (Figure 3).

The heterogeneity between studies was evaluated by the I^2 statistics. There was moderate heterogeneity among the results of the studies. There was no publication bias among studies confirmed by Begg and Egger tests (P=0.099 and P=0.280, respectively).

1 st author, y	Country	Design	Sample size	OR (Lower- Upper 95% Cl)	Estimate	Adjust- ment	Maternal age (y)	Quality of studies
Saadat, 2007 [25]	Iran	Case-control	250	0.77 (0.43, 1.38)	Odds ratio	Crude	Control group: 39.2±2 Case group: 37.2±2	Low
Ventura Lave- riano, 2013 [15]	Peru	Cohort	30935	1.40 (1.20, 1.60)	Odds ratio	Adjusted	Not reported	High
Tandu-Umba, 2014 <mark>[26]</mark>	Congo	Cross-sec- tional	2086	1.70 (1.10, 2.50)	Odds ratio	Adjusted	28.1	High
Cho, 2015 [20]	Korea	Cohort	222137	1.23 (1.08, 1.40)	Odds ratio	Adjusted	30.9	High
Dammavalam, 2017 [21]	India	Case-control	950	2.08 (1.14, 3.77)	Odds ratio	Crude	Not reported	High
Mbah, 2007 [24]	USA	Cohort	166712	1.28 (1.20, 1.37)	Odds ratio	Adjusted	Not reported	High
lyoke, 2014 [23]	Nigeria	Cohort	870	6.67 (1.50, 9.73)	Odds ratio	Crude	32.3	High

Table 1. Characteristics of included studies in the present meta-analysis

Discussion

The findings of this meta-analysis presented a significant association between previous cesarean section and the risk of preeclampsia. We reported a 28% increased risk for preeclampsia in women with previous cesarean section. The heterogeneity among studies was low. According to the NOS, the quality of studies was high, except for one study. The subgroup analysis was based on the adjusted form of studies. OR in crude and adjusted studies were reported at 1.29 and 1.29, respectively.

Preeclampsia leads to inefficient uteroplacental blood loss, negatively affecting perinatal outcomes [27]. A study

has reported several fetal complications from placental ischemia: intrauterine growth restriction (IUGR), oligohydramnios, fetal instability, and placental abruption. As a result, these complications increase the risk of premature delivery, whether spontaneous or indicated [28].

The association mechanism between previous cesarean section and the risk of preeclampsia is unclear. The initial pathology in preeclampsia is associated with maternal-fetal complications. It is determined by the invasion of poor trophoblastic of the uterine and changes in uteroplacental blood flow [16]. In addition, it has been reported that a previous cesarean was associated with unexplained stillbirth in a subsequent pregnancy. It has



Figure 1. Flow of Information through the different phases of the systematic review



Figure 2. Forest plot of the association between previous cesarean section and preeclampsia

been suggested that this association may be a manifestation of abnormal uterine blood flow caused by intentional or inadvertent occlusion of major uterine vessels during a previous cesarean and abnormal placentation from uterine scar [29]. risk factors that were the cause of cesarean in the first pregnancy, but not the cesarean itself, are attributed to preeclampsia in the next pregnancy. Furthermore, a higher recurrence risk of preeclampsia is known to be associated with earlier gestational age at delivery in a previous pregnancy complicated by preeclampsia [30].

Several factors (such as maternal obesity, diabetes mellitus, etc.) that are risk factors for preeclampsia are also risk factors for cesarean section. Therefore, these

These abnormalities may be ascribed to placenta ischemia that releases factors into the mother's circulation.

Study			%
ID		OR (95% CI)	Weight
	li		
Crude			
Saadat, 2007	•	0.77 (0.43, 1.38)	6.24
Dammavalam, 2017	-	2.08 (1.14, 3.77)	0.92
lyoke, 2014	· · · · · · · · · · · · · · · · · · ·	6.67 (1.50, 29.73)	0.01
Subtotal (I-squared = 50.1%, p = 0.135)	\diamond	1.29 (0.13, 2.46)	7.16
Adjusted			
Laveriano, 2013	•	1.40 (1.20, 1.60)	22.10
Tandu-umba, 2014		1.70 (1.10, 2.50)	3.08
Cho, 2015	•	1.23 (1.08, 1.40)	27.53
Mbah, 2007	•	1.28 (1.20, 1.37)	40.12
Subtotal (I-squared = 2.4%, p = 0.381)	1	1.29 (1.22, 1.36)	92.84
Overall (I-squared = 37.2%, p = 0.145)		1.28 (1.15, 1.41)	100.00
NOTE: Weights are from random effects analysis			

Figure 3. Forest plot of the association between previous cesarean section and preeclampsia based on the adjusted form of studies

These factors can show the clinical manifestations of the disease [31]. Thus, the changes in the uterus are due to surgery or manipulation of the uterus during cesarean section. This change may interfere with normal trophoblastic invasion and changes in placental uterine blood flow in subsequent pregnancies, leading to preeclampsia. On the other hand, the scar tissue of the cesarean section shows significant pathological changes such as lower uterine distortion, upper endometrial scarring, lymphocytic infiltration, the suture material of residual with cell reactions, capillary dilatation, fragmentation and breakdown of the endometrium. Also, biochemical behaviors show decreased levels of transforming factor beta3 and connective tissue growth factor but a slight increase in tumor necrosis factor [32].

The no publication bias and low heterogeneity are the strengths of this study. However, there were a few limitations in our study. We could not evaluate the effect of confounding variables in all studies. Three studies were reported in crude form. Therefore, this issue may raise the possibility of bias. Also, data extracted by the studies included in the meta-analysis were insufficient to conduct some subgroup analyses based on the number of previous cesarean sections. Therefore, we cannot subgroup analysis based on the number of previous cesarean sections and preeclampsia. However, the findings with 423940 participants and low heterogeneity showed that previous cesarean section is a risk factor for preeclampsia. This issue should be considered, especially for pregnant women who undergo cesarean delivery based on their request without any maternal or fetal indication.

In summary, these findings showed that previous cesarean section is a risk factor for preeclampsia. Therefore, programs and educational interventions should be considered to reduce elective cesarean section on maternal request.

Ethical Considerations

Compliance with ethical guidelines

This study was approved by Hamadan University of Medical Sciences (Project Code: 9811088568 and Ethical Code: IR.UMSHA.REC.1398.891).

Funding

This study was supported by Hamadan University of Medical Sciences (Grant No.: 9811088568).

Authors' contributions

Study design: Ensiyeh Jenabi and Soodabeh Aghababaei; Data conduction and data analysis: Salman Khazaei; Preaparing the manuscript: Ensiyeh Jenabi and Soodabeh Aghababaei; Final approval: All authors.

Conflict of interest

The authors declared no conflict of interest.

Acknowledgments

This study was financially supported by the Vice Chancellor for Research and Technology of Hamadan University of Medical Sciences.

References

- [1] Fujii T, Nagamatsu T, Morita K, Schust DJ, Iriyama T, Komatsu A, et al. Enhanced HIF2α expression during human trophoblast differentiation into syncytiotrophoblast suppresses transcription of placental growth factor. Scientific Reports. 2017; 7(1):12455. [DOI:10.1038/ s41598-017-12685-w] [PMID] [PMCID]
- [2] Gilbert JS, Gilbert SA, Arany M, Granger JP. Hypertension produced by placental ischemia in pregnant rats is associated with increased soluble endoglin expression. Hypertension. 2009; 53(2):399-403. [DOI:10.1161/HYPERTENSIONAHA.108.123513] [PMID] [PMCID]
- [3] Powe CE, Levine RJ, Karumanchi SA. Preeclampsia, a disease of the maternal endothelium: The role of antiangiogenic factors and implications for later cardiovascular disease. Circulation. 2011; 123(24):2856-69. [DOI:10.1161/CIRCULATIONAHA.109.853127]
 [PMID] [PMCID]
- [4] Steegers EA, von Dadelszen P, Duvekot JJ, Pijnenborg R. Pre-eclampsia. Lancet. 2010; 376(9741):631-44. [DOI:10.1016/S0140-6736(10)60279-6] [PMID]
- [5] Hypertension in pregnancy. Report of the American college of obstetricians and gynecologists' task force on hypertension in pregnancy. Obstetrics and Gynecology. 2013; 122(5):1122-31.
 [DOI:10.1097/01.AOG.0000437382.03963.88] [PMID]
- [6] Yang Y, Le Ray I, Zhu J, Zhang J, Hua J, Reilly M. Preeclampsia prevalence, risk factors, and pregnancy outcomes in Sweden and China. JAMA Network Open. 2021; 4(5):e218401. [DOI:10.1001/jamanetworkopen.2021.8401] [PMID] [PMCID]
- [7] Duckitt K, Harrington D. Risk factors for pre-eclampsia at antenatal booking: Systematic review of controlled studies. BMJ. 2005; 330(7491):565. [DOI:10.1136/bmj.38380.674340.E0] [PMID] [PMCID]

- [8] Lisonkova S, Joseph KS. Incidence of preeclampsia: Risk factors and outcomes associated with early- versus late-onset disease. American Journal of Obstetrics and Gynecology. 2013; 209(6):544.e1-12. [DOI:10.1016/j.ajog.2013.08.019] [PMID]
- [9] Tong W, Giussani DA. Preeclampsia link to gestational hypoxia. Journal of Developmental Origins of Health and Disease. 2019; 10(3):322-33. [DOI:10.1017/S204017441900014X] [PMID] [PMCID]
- [10] Veisani Y, Jenabi E, Delpisheh A, Khazaei S. Angiogenic factors and the risk of preeclampsia: A systematic review and meta-analysis. International Journal of Reproductive Biomedicine. 2019; 17(1):1–10. [DOI:10.18502/ijrm.v17i1.3815] [PMID] [PMCID]
- [11] Poorolajal J, Jenabi E. The association between body mass index and preeclampsia: A meta-analysis. The Journal of Maternal-Fetal & Neonatal Medicine. 2016; 29(22):3670-6. [DOI:10.3109/14767058. 2016.1140738] [PMID]
- [12] Patten IS, Rana S, Shahul S, Rowe GC, Jang C, Liu L, et al. Cardiac angiogenic imbalance leads to peripartum cardiomyopathy. Nature. 2012; 485(7398):333-8.[DOI:10.1038/nature11040] [PMID] [PM-CID]
- [13] Abenhaim HA, Benjamin A. Effect of prior cesarean delivery on neonatal outcomes. Journal of perinatal medicine. 2011; 39(3):241 4. [DOI:10.1515/jpm.2011.050]
- [14] Daltveit AK, Tollånes MC, Pihlstrøm H, Irgens LM. Cesarean delivery and subsequent pregnancies. Obstetrics and Gynecology. 2008; 111(6):1327-34. [DOI:10.1097/AOG.0b013e3181744110] [PMID]
- [15] Ventura Laveriano WR, Redondo CE. Obstetric outcomes in the second birth of women with a previous caesarean delivery: A retrospective cohort study from Peru. Revista Brasileira de Ginecologia e Obstetricia. 2013; 35(4):148-52. [DOI:10.1590/S0100-72032013000400003] [PMID]
- [16] Cross JC, Werb Z, Fisher SJ. Implantation and the placenta: Key pieces of the development puzzle. Science. 1994; 266(5190):1508-18. [DOI:10.1126/science.7985020] [PMID]
- [17] Hamilton BE, Hoyert DL, Martin JA, Strobino DM, Guyer B. Annual summary of vital statistics: 2010-2011. Pediatrics. 2013; 131(3):548-58. [DOI:10.1542/peds.2012-3769] [PMID] [PMCID]
- [18] Keag OE, Norman JE, Stock SJ. Long-term risks and benefits associated with cesarean delivery for mother, baby, and subsequent pregnancies: Systematic review and meta-analysis. Plos Medicine. 2018; 15(1):e1002494. [DOI:10.1371/journal.pmed.1002494]
 [PMID] [PMID]
- [19] Morris H. Surgical pathology of the lower uterine segment caesarean section scar: Is the scar a source of clinical symptoms? International Journal of Gynecological Pathology. 1995; 14(1):16-20. [DOI:10.1097/00004347-199501000-00004] [PMID]
- [20] Cho GJ, Kim LY, Min KJ, Sung YN, Hong SC, Oh MJ, et al. Prior cesarean section is associated with increased preeclampsia risk in a subsequent pregnancy. BMC Pregnancy and Childbirth. 2015; 15:24. [DOI:10.1186/s12884-015-0447-x] [PMID] [PMCID]
- [21] Dammavalam DS, Kushtagi P. Pregnancy-related hypertension in multigravidas with previous cesarean delivery. Istanbul Medical Journal. 2017; 18(2):80-5. [DOI:10.5152/imj.2017.70431]
- [22] Wells GA, Shea B, O'Connell D, Peterson J, Welch V, Losos M, et al. The newcastle-ottawa scale (NOS) for assessing the quality of nonrandomised studies in meta-analyses [Internet]. 2009 [Updated 2023 June 11]. Available from: [Link]

- [23] Iyoke CA, Ugwu GO, Ezugwu FO, Lawani OL, Onah HE. Risks associated with subsequent pregnancy after one caesarean section: A prospective cohort study in a Nigerian obstetric population. Nigerian Journal of Clinical Practice. 2014; 17(4):442-8. [DOI:10.4103/1119-3077.134035] [PMID]
- [24] Mbah AK, Sharma PP, Alio AP, Fombo DW, Bruder K, Salihu HM. Previous cesarean section, gestational age at first delivery and subsequent risk of pre-eclampsia in obese mothers. Archives of Gynecology and Obstetrics. 2012; 285(5):1375-81. [DOI:10.1007/ s00404-011-2161-x] [PMID]
- [25] Saadat M, Nejad SM, Habibi G, Sheikhvatan M. Maternal and neonatal outcomes in women with preeclampsia. Taiwanese Journal of Obstetrics & Gynecology. 2007; 46(3):255-9. [DOI:10.1016/S1028-4559(08)60029-7] [PMID]
- [26] Tandu-Umba B, Mbangama MA, Kamongola KM, Kamgang Tchawou AG, Kivuidi MP, et al. Pre-pregnancy high-risk factors at first antenatal visit: how predictive are these of pregnancy outcomes? International Journal of Women's Health. 2014; 6:1011-8. [DOI:10.2147/IJWH.S69230] [PMID] [PMCID]
- [27] Espinoza J. Uteroplacental ischemia in early- and late-onset preeclampsia: A role for the fetus? Ultrasound in Obstetrics & Gynecology. 2012; 40(4):373-82. [DOI:10.1002/uog.12280] [PMID]
- [28] ACOG Practice Bulletin No. 202: Gestational hypertension and preeclampsia. Obstetrics and Gynecology. 2019; 133(1):1. [DOI:10.1097/AOG.00000000003018] [PMID]
- [29] Smith GC, Pell JP, Dobbie R. Caesarean section and risk of unexplained stillbirth in subsequent pregnancy. Lancet. 2003; 362(9398):1779-84. [DOI:10.1016/S0140-6736(03)14896-9]
 [PMID]
- [30] Mostello D, Kallogjeri D, Tungsiripat R, Leet T. Recurrence of preeclampsia: Effects of gestational age at delivery of the first pregnancy, body mass index, paternity, and interval between births. American Journal of Obstetrics and Gynecology. 2008; 199(1):55. e1-7. [DOI:10.1016/j.ajog.2007.11.058] [PMID]
- [31] Chaiworapongsa T, Chaemsaithong P, Yeo L, Romero R. Preeclampsia part 1: Current understanding of its pathophysiology. Nature Reviews Nephrology. 2014; 10(8):466-80. [DOI:10.1038/ nrneph.2014.102] [PMID] [PMCID]
- [32] Pollio F, Staibano S, Mascolo M, Salvatore G, Persico F, De Falco M, et al. Uterine dehiscence in term pregnant patients with one previous cesarean delivery: Growth factor immunoexpression and collagen content in the scarred lower uterine segment. American journal of Obstetrics and Gynecology. 2006; 194(2):527-34. [DOI:10.1016/j. ajog.2005.07.048] [PMID]